Author’s response to reviews

Title: Prediction of grip and key pinch strength in 978 healthy subjects

Authors:

Felix Angst (fangst@vtxmail.ch)
Susann Drerup (sr.s@gmx.net)
Stephan Werle (stephan.werle@bluewin.ch)
Daniel D Herren (daniel.herren@kws.ch)
Beat R Simmen (beat.simmen@kws.ch)
Jörg Goldhahn (joerg.goldhahn@kws.ch)

Version: 6 Date: 11 May 2010

Author’s response to reviews:

Author list: Has been changed and should be correct now. If not indicate what exactly has to be corrected not only indicating a standard general statement...

References: have been changed into text format.

The appendix has been changed into a table so that lines are allowed.

Ethical approval: has been mentioned and referenced in Methods, Data collection